\nopagebreak
\begin{thebibliography}{99}
\frenchspacing
\newcommand{\epj}[3]{{\sl Eur. Phys. J.} {\bf #1} (19#2) #3}
\newcommand{\zp}[3]{{\sl Z. Phys.} {\bf #1} (19#2) #3}
\newcommand{\np}[3]{{\sl Nucl. Phys.} {\bf #1} (19#2) #3}
\newcommand{\phm}[3]{{\sl Phil. Mag.} {\bf #1} (19#2) #3}
\newcommand{\pl}[3]{{\sl Phys. Lett.} {\bf #1} (19#2) #3}
\newcommand{\pr}[3]{{\sl Phys. Rev.} {\bf #1} (19#2) #3}
\newcommand{\prep}[3]{{\sl Phys.\ Rep.} {\bf #1} (19#2) #3}
\newcommand{\prl}[3]{{\sl Phys. Rev. Lett.} {\bf #1} (19#2) #3}
\newcommand{\prs}[3]{{\sl Proc. Roy. Soc.} {\bf #1} (19#2) #3}
\newcommand{\fp}[3]{{\sl Fortschr. Phys.} {\bf #1} (19#2) #3}
\newcommand{\cpc}[3]{{\sl Comput. Phys. Commun.} {\bf #1} (19#2) #3}
\newcommand{\ijmp}[3]{{\sl Int. J. Mod. Phys.} {\bf #1} (19#2) #3}
\newcommand{\nim}[3]{{\sl Nucl. Instr. Meth.} {\bf #1} (19#2) #3}
\newcommand{\nc}[3]{{\sl Nuovo Cimento} {\bf #1} (19#2) #3}
\newcommand{\vj}[4]{{\sl #1} {\bf #2} (19#3) #4}
\newcommand{\jcp}[3]{{\sl J. Comp. Phys.} {\bf #1} (19#2) #3}

%\cite{ArkaniHamed:1998rs}
\bibitem{ArkaniHamed:1998rs}
  N.~Arkani-Hamed, S.~Dimopoulos and G.~R.~Dvali,
  %``The Hierarchy problem and new dimensions at a millimeter,''
  Phys.\ Lett.\  {\bf B429} (1998)  263-272.
  [hep-ph/9803315].

%\cite{Karg:2009xk}
\bibitem{Karg:2009xk}
  S.~Karg, M.~Kr\"amer, Q.~Li and D.~Zeppenfeld,
  %``NLO QCD corrections to graviton production at hadron colliders,''
  Phys.\ Rev.\ D {\bf 81} (2010) 094036
  [arXiv:0911.5095 [hep-ph]].
  %%CITATION = ARXIV:0911.5095;%%

%\cite{Karg:2010zz}
\bibitem{Karg:2010zz}
  S.~Karg  {\it et al.},
%  S.~Karg, T.~Binoth, T.~Gleisberg, N.~Kauer, G.~Sanguinetti, M.~Kr\"amer, Q.~Li and D.~Zeppenfeld,
  %``ZZ+jet and Graviton+jet at NLO QCD: recent applications using GOLEM methods,''
  PoS RADCOR {\bf 2009} (2010) 006.
  %%CITATION = POSCI,RADCOR2009,006;%%

%\cite{Georgi:2007ek}
\bibitem{Georgi:2007ek}
  H.~Georgi,
  %``Unparticle physics,''
  Phys.\ Rev.\ Lett.\  {\bf 98 } (2007)  221601.
  [hep-ph/0703260].
  
%\cite{Betz:2006ds}
\bibitem{Betz:2006ds}
%  B.~Betz, M.~Bleicher, U.~Harbach, T.~Humanic, B.~Koch and H.~Stoecker,
B.~Betz {\it et al.},
  %``Mini black holes at the LHC: Discovery through di-jet suppression, mono-jet emission and a supersonic boom in the quark-gluon plasma in ALICE, ATLAS and CMS,''
  hep-ph/0606193.
  %%CITATION = HEP-PH/0606193;%%

%\cite{Rizzo:2008fp}
\bibitem{Rizzo:2008fp}
  T.~G.~Rizzo,
  %``Identification of the Origin of Monojet Signatures at the LHC,''
  Phys.\ Lett.\  {\bf B665} (2008)  361-368.
  [arXiv:0805.0281 [hep-ph]].

%\cite{Aad:2011xw}
\bibitem{Aad:2011xw}
  G.~Aad {\it et al.}  [ATLAS Collaboration],
  %``Search for new phenomena with the monojet and missing transverse momentum signature using the ATLAS detector in sqrt(s) = 7 TeV proton-proton collisions,''
  Phys.\ Lett.\ B {\bf 705} (2011) 294
  [arXiv:1106.5327 [hep-ex]].
  %%CITATION = ARXIV:1106.5327;%%

%\mpar{TK: references combined}
%\cite{Chatrchyan:2011nd}
\bibitem{Chatrchyan:2011nd}
  S.~Chatrchyan {\it et al.}  [CMS Collaboration],
  %``Search for New Physics with a Mono-Jet and Missing Transverse Energy in $pp$ Collisions at $\sqrt{s} = 7$ TeV,''
  Phys.\ Rev.\ Lett.\  {\bf 107} (2011) 201804
  [arXiv:1106.4775 [hep-ex]] and 
  %%CITATION = ARXIV:1106.4775;%%
%\cite{Chatrchyan:2012me}
%\bibitem{Chatrchyan:2012me}
%  S.~Chatrchyan {\it et al.}  [CMS Collaboration],
  %``Search for dark matter and large extra dimensions in monojet events in $pp$ collisions at $\sqrt{s}=7$ TeV,''
  JHEP {\bf 1209} (2012) 094
  [arXiv:1206.5663 [hep-ex]].
  %%CITATION = ARXIV:1206.5663;%%

%\cite{Vacavant:2001sd}
\bibitem{Vacavant:2001sd}
  L.~Vacavant and I.~Hinchliffe,
  %``Signals of models with large extra dimensions in ATLAS,''
  J.\ Phys.\ G {\bf G27} (2001)  1839-1850.

%\cite{Benucci:2010ax}
\bibitem{Benucci:2010ax}
  L.~Benucci, [for the CMS~Collaboration],
  %``Search for Extra dimensions in a single-jet and missing energy channel at CMS experiment,''
  PoS {\bf HCP2009} (2009)  065.
  [arXiv:1001.5428 [physics.ins-det]].

%%\cite{deFlorian:2010zn}
%\bibitem{deFlorian:2010zn}
%  D.~de Florian, F.~Wagner,
%  %``Single-inclusive jet production in polarized pp collisions at RHIC to next-to-leading% logarithmic accuracy,''
%  Eur.\ Phys.\ J.\  {\bf C67} (2010)  367-375.
%  [arXiv:1004.0764 [hep-ph]].

%\cite{Giele:1993dj}
\bibitem{Giele:1993dj}
  W.~T.~Giele, E.~W.~N.~Glover and D.~A.~Kosower,
  %``Higher order corrections to jet cross-sections in hadron colliders,''
  Nucl.\ Phys.\  B {\bf 403} (1993) 633
  [hep-ph/9302225].
  %%CITATION = NUPHA,B403,633;%%
  
 %\cite{Campbell:2002tg}
\bibitem{Campbell:2002tg}
  J.~M.~Campbell and R.~K.~Ellis,
  %``Next-to-leading order corrections to $W^+$ 2 jet and $Z^+$ 2 jet production
  %at hadron colliders,''
  Phys.\ Rev.\  D {\bf 65} (2002) 113007
  [hep-ph/0202176].
  %%CITATION = PHRVA,D65,113007;%% 

%\cite{vanderBij:1988ac}
\bibitem{vanderBij:1988ac}
 J.~J.~van der Bij and E.~W.~N.~Glover,
 %``Z Boson Production And Decay Via Gluons,''
 Nucl.\ Phys.\ B {\bf 313} (1989)  237. 


%\cite{Alioli:2010qp}
\bibitem{Alioli:2010qp}
  S.~Alioli, P.~Nason, C.~Oleari and E.~Re,
  %``Vector boson plus one jet production in POWHEG,''
  JHEP {\bf 1101} (2011)  095
  [arXiv:1009.5594 [hep-ph]]. 

%\cite{Becher:2011fc}
\bibitem{Becher:2011fc}
  T.~Becher, C.~Lorentzen and M.~D.~Schwartz,
  %``Resummation for W and Z production at large pT,''
  Phys.\ Rev.\ Lett.\  {\bf 108} (2012) 012001
  [arXiv:1106.4310 [hep-ph]].
  %%CITATION = ARXIV:1106.4310;%%

%\cite{Kuhn:2004em}
\bibitem{Kuhn:2004em}
  J.~H.~K\"uhn, A.~Kulesza, S.~Pozzorini and M.~Schulze,
  %``Logarithmic electroweak corrections to hadronic Z+1 jet production at large transverse momentum,''
  Phys.\ Lett.\  B {\bf 609} (2005)  277
  [hep-ph/0408308].

%\cite{Kuhn:2005az}
\bibitem{Kuhn:2005az}
  J.~H.~K\"uhn, A.~Kulesza, S.~Pozzorini and M.~Schulze,
  %``One-loop weak corrections to hadronic production of Z bosons at large transverse momenta,''
  Nucl.\ Phys.\  B {\bf 727} (2005) 368
  [hep-ph/0507178].

%\cite{Ciafaloni:1998xg}
\bibitem{Ciafaloni:1998xg}
  P.~Ciafaloni and D.~Comelli,
  %``Sudakov enhancement of electroweak corrections,''
  Phys.\ Lett.\  B {\bf 446} (1999) 278
  [hep-ph/9809321];\\
  %%CITATION = PHLTA,B446,278;%%
%\bibitem{Fadin:1999bq}
  V.~S.~Fadin, L.~N.~Lipatov, A.~D.~Martin and M.~Melles,
  %``Resummation of double logarithms in electroweak high energy processes,''
  Phys.\ Rev.\  D {\bf 61} (2000) 094002
  [hep-ph/9910338];\\
  %%CITATION = PHRVA,D61,094002;%%
%
%\cite{Denner:2000jv}
%\bibitem{Denner:2000jv}
  A.~Denner and S.~Pozzorini,
  %``One-loop leading logarithms in electroweak radiative corrections. I:
  %Results,''
  Eur.\ Phys.\ J.\  C {\bf 18} (2001) 461
  [hep-ph/0010201];\\
  %%CITATION = EPHJA,C18,461;%%
%
%\bibitem{Beenakker:2001kf}
  W.~Beenakker and A.~Werthenbach,
  %``Electroweak two-loop Sudakov logarithms for on-shell fermions and
  %bosons,''
  Nucl.\ Phys.\  B {\bf 630} (2002) 3
  [hep-ph/0112030];\\
  %%CITATION = NUPHA,B630,3;%%
%
%\bibitem{Denner:2003wi}
  A.~Denner, M.~Melles and S.~Pozzorini,
  %``Two-loop electroweak angular-dependent logarithms at high energies,''
  Nucl.\ Phys.\  B {\bf 662} (2003) 299
  [hep-ph/0301241].
  %%CITATION = NUPHA,B662,299;%%

%\cite{Aad:2011qv}
\bibitem{Aad:2011qv}
  G.~Aad {\it et al.}  [ATLAS Collaboration],
  %``Measurement of the production cross section for Z/gamma* in association with jets in $pp$ collisions at $\sqrt{s}=7$ TeV with the ATLAS detector,''
  Phys.\ Rev.\ D {\bf 85} (2012) 032009
  [arXiv:1111.2690 [hep-ex]].
  %%CITATION = ARXIV:1111.2690;%%

%\cite{Chatrchyan:2011wt}
\bibitem{Chatrchyan:2011wt}
  S.~Chatrchyan {\it et al.}  [CMS Collaboration],
  %``Measurement of the Rapidity and Transverse Momentum Distributions of $Z$ Bosons in $pp$ Collisions at $\sqrt{s}=7$ TeV,''
  Phys.\ Rev.\ D {\bf 85} (2012) 032002
  [arXiv:1110.4973 [hep-ex]].
  %%CITATION = ARXIV:1110.4973;%%

%\cite{Ita:2011wn}
\bibitem{Ita:2011wn}
  H.~Ita  {\it et al.},
  %``Precise Predictions for Z + 4 Jets at Hadron Colliders,''
  Phys.\ Rev.\ D {\bf 85} (2012) 031501
  [arXiv:1108.2229 [hep-ph]].
  %%CITATION = ARXIV:1108.2229;%%


%\cite{Watt:2011kp}
\bibitem{Watt:2011kp}
  G.~Watt,
  %``Parton distribution function dependence of benchmark Standard Model total cross sections at the 7 TeV LHC,''
  JHEP {\bf 1109} (2011) 069
  [arXiv:1106.5788 [hep-ph]].
  %%CITATION = ARXIV:1106.5788;%%

\bibitem{Denner:2005fg}
  A.~Denner, S.~Dittmaier, M.~Roth and L.~H.~Wieders,
  %``Electroweak corrections to charged-current e+ e- $\to$ 4 fermion processes:
  %Technical details and further results,''
  Nucl.\ Phys.\  B {\bf 724} (2005) 247
  [hep-ph/0505042];\\
%
%\bibitem{Denner:2006ic}
  A.~Denner and S.~Dittmaier,
  % ``The complex-mass scheme for perturbative calculations with unstable
  %particles,''
  Nucl.\ Phys.\ Proc.\ Suppl.\  {\bf 160} (2006) 22
  [hep-ph/0605312].
  %%CITATION = HEP-PH 0505042;%%

%\cite{Denner:2011vu}
\bibitem{Denner:2011vu}
  A.~Denner, S.~Dittmaier, T.~Kasprzik and A.~M\"uck,
  %``Electroweak corrections to dilepton + jet production at hadron colliders,''
  JHEP {\bf 1106} (2011)  069.
  [arXiv:1103.0914 [hep-ph]].
\bibitem{Martin:2004dh}
  A.~D.~Martin, R.~G.~Roberts, W.~J.~Stirling and R.~S.~Thorne,
  %``Parton distributions incorporating QED contributions,''
  Eur.\ Phys.\ J.\ C {\bf 39} (2005) 155
  [hep-ph/0411040].
  %%CITATION = HEP-PH 0411040;%%

\bibitem{Kublbeck:1990xc}
  J.~K\"ublbeck, M.~B\"ohm and A.~Denner,
  %``Feyn Arts: Computer Algebraic Generation Of Feynman Graphs And Amplitudes,''
  Comput.\ Phys.\ Commun.\  {\bf 60} (1990) 165;\\
  %%CITATION = CPHCB,60,165;%%
  H.~Eck and J.~K\"ublbeck, {\it Guide to FeynArts 1.0\/},
  University of W\"urzburg, 1992;\\
%\bibitem{Hahn:2000kx}
  T.~Hahn,
  %``Generating Feynman diagrams and amplitudes with FeynArts 3,''
  Comput.\ Phys.\ Commun.\  {\bf 140} (2001) 418
  [hep-ph/0012260].

%\cite{Accomando:2005ra}
\bibitem{Accomando:2005ra}
  E.~Accomando, A.~Denner and C.~Meier,
  %``Electroweak corrections to $W \gamma$ and $Z \gamma$ production at the
  %LHC,''
  Eur.\ Phys.\ J.\  C {\bf 47} (2006) 125
  [hep-ph/0509234].
  %%CITATION = EPHJA,C47,125;%%
\bibitem{Hahn:1998yk}
  T.~Hahn and M.~P\'erez-Victoria,
  %``Automatized one-loop calculations in four and D dimensions,''
  Comput.\ Phys.\ Commun.\  {\bf 118} (1999) 153
  [hep-ph/9807565].
  %%CITATION = HEP-PH 9807565;%%

\bibitem{Denner:2009gj}
  A.~Denner, S.~Dittmaier, T.~Kasprzik and A.~M\"uck,
  %``Electroweak corrections to W + jet hadroproduction including leptonic W-boson decays,''
  JHEP {\bf 0908} (2009)  075.
  [arXiv:0906.1656 [hep-ph]].
%\cite{Alwall:2007st}
%\bibitem{Alwall:2007st}
%  J.~Alwall {\it et al.},
%  %``MadGraph/MadEvent v4: The New Web Generation,''
%  JHEP {\bf 0709} (2007)  028.
%  [arXiv:0706.2334 [hep-ph]].

\bibitem{Passarino:1979jh}
  G.~Passarino and M.~Veltman,
  %``One Loop Corrections For E+ E- Annihilation Into Mu+ Mu- In The Weinberg Model,''
  Nucl.\ Phys.\ B {\bf 160} (1979) 151;\\
  %%CITATION = NUPHA,B160,151;%%
%\cite{Denner:2002ii}
%
%
%\bibitem{Denner:2002ii}
  A.~Denner and S.~Dittmaier,
  %``Reduction of one-loop tensor 5-point integrals,''
  Nucl.\ Phys.\ B {\bf 658} (2003) 175
  [hep-ph/0212259] and 
  %%CITATION = HEP-PH 0212259;%%
%
%\bibitem{Denner:2005nn}
%  A.~Denner and S.~Dittmaier,
  %``Reduction schemes for one-loop tensor integrals,''
  Nucl.\ Phys.\ B {\bf 734} (2006) 62
  [hep-ph/0509141].
  %%CITATION = HEP-PH 0509141;%%

\bibitem{'tHooft:1979xw}
  G.~'t Hooft and M.~Veltman,
  %``Scalar One Loop Integrals,''
  Nucl.\ Phys.\ B {\bf 153} (1979) 365;\\
  %%CITATION = NUPHA,B153,365;%%
  %
  %\cite{Beenakker:1990jr}
  %\bibitem{Beenakker:1990jr}
  W.~Beenakker and A.~Denner,
  %``Infrared Divergent Scalar Box Integrals With Applications In The Electroweak Standard Model,''
  Nucl.\ Phys.\ B {\bf 338} (1990) 349;\\
  %%CITATION = NUPHA,B338,349;%%
  %
%\cite{Denner:2010tr}
%\bibitem{Denner:2010tr}
  A.~Denner and S.~Dittmaier,
  %``Scalar one-loop 4-point integrals,''
  Nucl.\ Phys.\  B {\bf 844} (2011) 199
  [arXiv:1005.2076 [hep-ph]].
  %%CITATION = NUPHA,B844,199;%%
\bibitem{Dittmaier:1999mb}
  S.~Dittmaier,
  %``A general approach to photon radiation off fermions,''
  Nucl.\ Phys.\ B {\bf 565} (2000) 69
  [hep-ph/9904440].
  %%CITATION = HEP-PH 9904440;%%
%\cite{Dittmaier:2008md}
\bibitem{Dittmaier:2008md}
  S.~Dittmaier, A.~Kabelschacht and T.~Kasprzik,
  %``Polarized QED splittings of massive fermions and dipole subtraction for
  %non-collinear-safe observables,''
  Nucl.\ Phys.\  B {\bf 800} (2008) 146
  [arXiv:0802.1405 [hep-ph]].
  %%CITATION = NUPHA,B800,146;%%

%\cite{Denner:2010ia}
\bibitem{Denner:2010ia}
  A.~Denner, S.~Dittmaier, T.~Gehrmann and C.~Kurz,
  %``Electroweak corrections to hadronic event shapes and jet production in e+e- annihilation,''
  Nucl.\ Phys.\  {\bf B836} (2010)  37 
  [arXiv:1003.0986 [hep-ph]].%
%\cite{Denner:2009gj}



\bibitem{Glover:1993xc}
  E.~W.~N.~Glover and A.~G.~Morgan,
  %``Measuring the photon fragmentation function at LEP,''
  Z.\ Phys.\  C {\bf 62} (1994) 311.
  %%CITATION = ZEPYA,C62,311;%%
\bibitem{Buskulic:1995au}
  D.~Buskulic {\it et al.}  [ALEPH Collaboration],
  %``First measurement of the quark to photon fragmentation function,''
  Z.\ Phys.\  C {\bf 69} (1996) 365.
  %%CITATION = ZEPYA,C69,365;%%
\bibitem{Amsler:2008zzb}
  C.~Amsler {\it et al.}  [Particle Data Group],
  %``Review of particle physics,''
  Phys.\ Lett.\  B {\bf 667} (2008) 1.
  %%CITATION = PHLTA,B667,1;%%
%\cite{Bardin:1988xt}
\bibitem{Bardin:1988xt}
  D.~Y.~Bardin, A.~Leike, T.~Riemann and M.~Sachwitz,
  %``Energy dependent width effects in $e^+e^-$ annihilation  near the $Z$
  %pole,''
  Phys.\ Lett.\  B {\bf 206} (1988) 539.
  %%CITATION = PHLTA,B206,539;%%
%\cite{Martin:2009iq}
\bibitem{Martin:2009iq}
  A.~D.~Martin {\it et al.},
  %``Parton distributions for the LHC,''
  Eur.\ Phys.\ J.\  C {\bf 63}, (2009) 189
  [arXiv:0901.0002 [hep-ph]].

%\cite{Whalley:2005nh}
\bibitem{Whalley:2005nh}
  M.~R.~Whalley, D.~Bourilkov and R.~C.~Group, in {\sl HERA and the
    LHC}, eds.\ A. de Roeck and H. Jung (CERN-2005-014, Geneva, 2005), p.~575,
  %``The Les Houches Accord PDFs (LHAPDF) and Lhaglue,''
  hep-ph/0508110.
  %%CITATION = HEP-PH/0508110;%%
%\cite{Bauer:2009km}
\bibitem{Bauer:2009km}
  C.~W.~Bauer and B.~O.~Lange,
  %``Scale setting and resummation of logarithms in pp -> V + jets,''
  arXiv:0905.4739 [hep-ph].
  %%CITATION = ARXIV:0905.4739;%%
%\cite{Blazey:2000qt}
\bibitem{Blazey:2000qt}
  G.~C.~Blazey {\it et al.},
  in {\sl QCD and Weak Boson Physics in Run II}, eds.\ U. Baur, R.K. Ellis and D. Zeppenfeld
  (Fermilab-Pub-00/297, Fermilab, 2000) p.~47,
  %``Run II jet physics,''
  hep-ex/0005012.
  %%CITATION = HEP-EX/0005012;%%
%\cite{Cacciari:2008gp}
\bibitem{Cacciari:2008gp}
  M.~Cacciari, G.~P.~Salam and G.~Soyez,
  %``The Anti-k(t) jet clustering algorithm,''
  JHEP {\bf 0804} (2008) 063
  [arXiv:0802.1189 [hep-ph]].
  %%CITATION = ARXIV:0802.1189;%%
%\cite{Kuhn:2005gv}
\bibitem{Kuhn:2005gv}
  J.~H.~K\"uhn, A.~Kulesza, S.~Pozzorini and M.~Schulze,
  %``Electroweak corrections to hadronic photon production at large transverse momenta,''
  JHEP {\bf 0603} (2006) 059
  [hep-ph/0508253].
  %%CITATION = HEP-PH/0508253;%%
%\cite{GehrmannDeRidder:2012ja}
\bibitem{GehrmannDeRidder:2012ja}
  A.~Gehrmann-De Ridder, T.~Gehrmann and M.~Ritzmann,
  %``Antenna subtraction at NNLO with hadronic initial states: double real initial-initial configurations,''
  JHEP {\bf 1210} (2012) 047
  [arXiv:1207.5779 [hep-ph]];\\
  %%CITATION = ARXIV:1207.5779;%%
 T.~Gehrmann and L.~Tancredi,
  %``Two-loop QCD helicity amplitudes for $q\bar q \to W^\pm \gamma$ and $q\bar q \to Z^0 \gamma$,''
  JHEP {\bf 1202} (2012) 004
  [arXiv:1112.1531];\\
  %%CITATION = ARXIV:1112.1531;%%
%\cite{Daleo:2006xa}
%\bibitem{Daleo:2006xa}
  A.~Daleo, T.~Gehrmann and D.~Maitre,
  %``Antenna subtraction with hadronic initial states,''
  JHEP {\bf 0704} (2007) 016
  [hep-ph/0612257];\\
  %%CITATION = HEP-PH/0612257;%%
%\mpar{TK: references [60] from arXiv:1207.5779 added}
 L.~W.~Garland {\it et al.},
  %``The Two loop QCD matrix element for e+ e- ---> 3 jets,''
  Nucl.\ Phys.\ B {\bf 627} (2002) 107
  [hep-ph/0112081] and 
  %%CITATION = HEP-PH/0112081;%%
 %``Two loop QCD helicity amplitudes for e+ e- ---> three jets,''
  Nucl.\ Phys.\ B {\bf 642} (2002) 227
  [hep-ph/0206067].
  %%CITATION = HEP-PH/0206067;%%
\end{thebibliography}
